Author's response to reviews

Title: Impact of on-site cardiac catheterization on resource utilization and fatal and non-fatal outcomes after acute myocardial infarction

Authors:

Abdul R Halabi (abdul.halabi@duke.edu)
Christine A Beck (cabeck@dal.ca)
Mark J Eisenberg (meisenberg@epid.jgh.mcgill.ca)
Hugues Richard (Hugues@epimgh.mcgill.ca)
Louise Pilote (louise.pilote@mcgill.ca)

Version: 2 Date: 31 July 2006

Author's response to reviews: see over
Dear Dr. Norton:

Please find attached our manuscript entitled “Impact of on-site cardiac catheterization on resource utilization and fatal and non-fatal outcomes after acute myocardial infarction”, which we are re-submitting to Health Services Research.

This manuscript represents original work and is not under consideration for publication elsewhere. The authors meet criteria for authorship: they have participated in the writing, conception and design of the study, and the analysis of the data of the manuscript. The final submitted version of the paper has been seen and approved by each author.

All corresponding mail should be addressed to Dr. Louise Pilote, Division of Clinical Epidemiology, Montreal General Hospital, 1650 Cedar Avenue, Suite L10-421, Montreal, Quebec, H3G 1A4. Tel.: (514) 934-1934 ext. 44722, Fax: (514) 934-8293, e-mail: louise.pilote@mcgill.ca

Sincerely,

Louise Pilote, MD, MPH, PhD
Associate Professor of Medicine, McGill University
Director, Division of General Internal Medicine
McGill University Health Centre and McGill University

Encl.